Author's response to reviews

Title: Tumor-suppressor activity of RRIG1 in breast cancer

Authors:

Guihong Zhang (zgh7474@yahoo.com.cn)
Abenaa Brewster (ABrewster@mdanderson.org)
Baoxiang Guan (bguan@mdanderson.org)
Zhen Fan (zfan@mdanderson.org)
Powel H. Brown (pbrown@mdanderson.org)
Xiao-Chun Xu (xxu@mdanderson.org)

Version: 2 Date: 18 October 2010

Author's response to reviews:

This ms was submitted to Breast Cancer Res. The reviewers rejected our manuscript based on our data on the specificity of our antibody used in the manuscript. For the new submission, we have fixed this problem by added more experiments to have verified the specificity of the RRIG1 antibody; so I would like you to reconsider this manuscript for publication in BMC Cancer, which was introduced by Philippa.

Many thanks for your consideration and we are looking forward to hearing from you.